Author's response to reviews

Title: Inguinal lymph node metastases from a testicular seminoma a rare occurrence: A case report and review of literature

Authors:

Mohamed Ismail Jr (ms18273@gmail.com)
Farouq Zaman Jr (zaman17@hotmail.com)
Sohail Baithun Sr (s.i.baithun@gmul.ac.uk)
Vinod Nargund Sr (vinod.nargund@homerton.nhs.uk)
Jhumur Pati Sr (jhumur.pati@homerton.nhs.uk)
Junaid Masood Sr (junaid.masood@bartsandthelondon.nhs.uk)

Version: 4 Date: 4 October 2010

Author's response to reviews:

Dear Sirs

please find our revised manuscript, the following changes has been made

1- title corrected according to guidelines
2- patient identification in stated consistently
3- abstract in the submission page has been changed
4- competing interest section moved forward
5- initials added to the email addresses
6- manuscript checked for Typography

thanks

Yours sincerely

M. Ismail